Author’s response to reviews

Title: Absolute and estimated values of macular pigment optical density in young and aged Asian participants with or without age-related macular degeneration

Authors:

Yoko Ozawa (ozawa@a5.keio.jp)
Yuta Shigeno (yuta0shigeno@yahoo.co.jp)
Norihiro Nagai (nagai@a5.keio.jp)
Misa Suzuki (misayutakatomo@icloud.com)
Toshihide Kurihara (kuriboh2@yahoo.co.jp)
Sakiko Minami (saki.love5@icloud.com)
Eri Hirano (eriii_nn@yahoo.co.jp)
Hajime Shinoda (shinoha@mac.com)
Saori Kobayashi (saori_kobayashi@blueberryeye.co.jp)
Kazuo Tsubota (tsubota@z3.keio.jp)

Version: 1 Date: 03 Feb 2017

Author’s response to reviews:

Responses to the Editor’s comments

We appreciate the Editor’s assistance to revise the manuscript appropriately.

Editor's comments:

1) In the Declaration section, under Competing Interests, you state that the study was supported by Wakasa Seikatsu Co. Ltd. Can you please include in the Competing Interests that your co-author, Ms. Saori Kobayashi, works for this company? In addition, what involvement do Dr Yoko Ozawa and Dr. Kazuo Tsubota have with the company? For more information on our
policies regarding competing interests, please go to: http://www.biomedcentral.com/getpublished/editorial-policies#competing+interests

We included the below description in the Competing Interests part.

Competing interests

The MPSII® was generously rent to us by M.E. Technica Co. Ltd. The study was supported by Wakasa Seikatsu Co. Ltd. SK is an employee of Wakasa Seikatsu Co. Ltd., and YO and KT received the grant from Wakasa Seikatsu Co. Ltd.

2) As there is no individual person's data in this study, your Consent to publish statement should state 'Not applicable'.

We revised into “Not applicable” in the Consent to publish statement part.

3) For figures 1 and 2, please remove the figure titles (Figure 1 and Figure 2) from the graphic files, as these should only be included in the main manuscript.

We deleted the titles in the figures 1 and 2.